

# Incident atrial fibrillation and adverse clinical outcomes during extended follow-up of participants recruited to the remote heart rhythm sampling using the AliveCor heart monitor to screen for atrial fibrillation: the REHEARSE-AF study

Elizabeth A. Ellins (1,\*, Kathie Wareham<sup>2</sup>, Daniel E. Harris<sup>1,3</sup>, Matthew Hanney<sup>2</sup>, Ashley Akbari (1, Mark Gilmore<sup>4</sup>, James P. Barry<sup>5</sup>, Ceri J. Phillips<sup>6</sup>, Michael B. Gravenor<sup>1</sup>, and Julian P. Halcox<sup>1</sup>

<sup>1</sup>Population Data Science, Swansea University Medical School, Faculty of Medicine, Health & Life Science, Swansea University, Singleton Park, Swansea, SA2 8PP, UK; <sup>2</sup>Faculty of Medicine, Health & Life Science, Swansea University Health Board, Llanelli, UK; <sup>4</sup>Cardiology, Princess of Wales Hospital, Bridgend, UK; <sup>5</sup>Regional Cardiac Centre, Morriston Hospital, Swansea, UK; and <sup>6</sup>Swansea University College of Health and Human Sciences, Faculty of Medicine, Health & Life Science, Swansea University, Swansea, UK

Received 7 February 2023; revised 25 April 2023; accepted 2 May 2023; online publish-ahead-of-print 3 May 2023

Handling Editor: Magnus Bäck

Aims	Atrial fibrillation (AF) is an important risk factor for stroke, which is commonly asymptomatic, particularly in older patients, and often undetected until cardiovascular events occur. Development of novel technology has helped to improve detection of AF. However, the longer-term benefit of systematic electrocardiogram (ECG) screening on cardiovascular outcomes is unclear.
Methods and results	In the original REHEARSE-AF study, patients were randomized to twice-weekly portable electrocardiogram (iECG) assessment or routine care. After discontinuing the trial portable iECG assessment, electronic health record data sources provided longer-term follow-up analysis. Cox regression was used to provide unadjusted and adjusted hazard ratios (HR) [95% confidence intervals (Cl)] for clinical diagnosis, events, and anticoagulant prescriptions during the follow-up period. Over the median 4.2-year follow-up, although a greater number of patients were diagnosed with AF in the original iECG group (43 vs. 31), this was not significant (HR 1.37, 95% Cl 0.86–2.19). No differences were seen in the number of strokes/systemic embolisms or deaths between the two groups (HR 0.92, 95% Cl 0.54–1.54; HR 1.07, 95% Cl 0.66–1.73). Findings were similar when restricted to those with CHADS-VASc $\geq$ 4.
Conclusion	A 1-year period of home-based, twice-weekly screening for AF increased diagnoses of AF for the screening period but did not lead to increased diagnoses of AF or a reduction in cardiovascular-related events or all-cause death over a median of 4.2 years, even in those at highest risk of AF. These results suggest that benefits of regular ECG screening over a 1-year period are not maintained after cessation of the screening protocol.

\* Corresponding author. Tel: +44 (0)1792 295092, Email: E.A.Ellins@swansea.ac.uk

© The Author(s) 2023. Published by Oxford University Press on behalf of the European Society of Cardiology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com

#### **Graphical Abstract**



#### Lay summary

A follow-up of patients who had undergone a 1-year screening programme to detect atrial fibrillation (AF) (a heart rhythm problem) was carried out using electronic health records to see if there were differences in AF diagnoses, strokes, or deaths between those who had the screening device and those who had routine care (RC).

- Extended follow-up of the patients showed no overall increase in the diagnosis of AF in the screened population, despite a higher detection rate during the screening period.
- Numbers of strokes/systemic embolisms and deaths were similar in the screened and RC groups over the extended period of follow-up (including the year of screening).

## Introduction

Atrial fibrillation (AF) is an important risk factor for stroke, which is commonly asymptomatic, particularly in older patients, and therefore often undetected until diagnosis at the time of an associated cardiovascular event.<sup>1,2</sup> However, identification of AF and subsequent treatment with oral anticoagulants reduce the risk of stroke and other cardiovascular events.<sup>3,4</sup> Development of novel devices and technology allowing easy and accurate electrocardiographic rhythm assessment has facilitated screening of patients, improving identification of those with AF. Indeed, a study (REHEARSE-AF) by our group in patients identified as being at high risk of AF demonstrated that twice-weekly electrocardiogram (ECG) screening with a WiFi-enabled iPod ECG device (iECG) identified significantly more incident AF over a 1-year period than those patients who received normal care.<sup>5</sup> Other studies have also shown that detection of AF can be increased using alternative screening strategies.<sup>6,7</sup>

However, the limited follow-up period of these studies has not allowed a fuller evaluation of the effectiveness of a 1-year screening intervention for preventing cardiovascular events over the longer term, beyond the screening period. Two recent studies, the STROKESTOP and LOOP studies, have looked at the effects of screening on the risk of stroke and other cardiovascular outcomes but with conflicting findings.<sup>8,9</sup>

Linking data from the REHEARSE-AF study to routinely held clinical data for the trial participants, this analysis aimed to explore potential differences in AF diagnoses, strokes, and death rates during the longer-term, routine clinical follow-up in patients who received routine care (RC) vs. those who underwent the twice-weekly iECG monitoring regime for the first year of the extended evaluation period.<sup>5</sup>

## Methods

#### Transparency and openness promotion

The data used in this evaluation are openly available in the Secure Anonymised Information Linkage (SAIL) Databank at Swansea University, Swansea, UK. Due to the sensitive nature of these data, all proposals to use SAIL data are subject to review by an independent Information Governance Review Panel (IGRP). SAIL has an established application process for all projects and users who want to access data via SAIL https://www. saildatabank.com/application-process. This project was approved by the IGRP at Swansea University (SAIL project number 0982).

#### **Original study**

The method of the original study is reported in Halcox et al.<sup>5</sup> Briefly, in 2015, 1001 patients (53.3% female) over 65 years (mean age 72.6 ± 5.4 years), with a CHADS-VASc score  $\geq$  2, without a known diagnosis of AF, known contraindication to anticoagulation or permanent cardiac pacing implantation, were recruited and allocated to either RC (n = 501) or the intervention arm (iECG, n = 500). Participants in the intervention arm undertook twice-weekly recording of a 30 s single-lead iECG trace for 12 months (AliveCor Kardia Mobile system, AliveCor Inc., Mountain View, CA, USA) attached to an iPod (Apple Inc., Cupertino, CA, USA), which was transmitted via an encrypted process to a secure server. Additional traces could be submitted if participants were symptomatic. The iECG traces were analysed by an automated analysis software algorithm [AliveCor version 2.2.0 (build 21)] and transmitted over a secure server for analysis by a physiologist-led electrocardiographic reading service (Technomed Ltd UK). Abnormal and 10% of normal ECGs were overread by a cardiologist. Clinical review and appropriate care were arranged for those with clinically significant arrhythmia. Patients in the RC arm were followed up as normal by their general practitioner (GP). Ethics approval was obtained from the Wales Research Ethics Committee 6 (REC reference 14/ WA/1227), and the clinical trial was registered at URL: https://www.isrctn. com (unique identifier: ISRCTN10709813).

# Table 1 Baseline characteristics of the study participants with data linkage

	iECG (n = 499)	Routine care (n = 498)
Age (SD) years	72.60 (5.5)	72.55 (5.4)
Female, n (%)	260 (52.1)	273 (54.8)
Heart failure, n (%)	5 (1.0)	9 (1.8)
Hypertension, n (%)	267 (53.7)	271 (55.0)
Diabetes mellitus, n (%)	128 (25.7)	140 (28.2)
Stroke, n (%)	35 (7.1)	28 (5.7)
Vascular disease, n (%)	71 (14.4)	78 (15.8)
CHADS-VASc score (SD)	2.97 (1.0)	3.01 (1.0)
Mean follow-up period (SD)	1534 (214)	1530 (222)

Vascular disease, ischaemic heart disease, or peripheral vascular disease.

#### Follow-up

The original study dataset was imported into the SAIL Databank, a world-leading privacy-protecting trusted research environment (TRE) that holds anonymized individual-level, population-scale data, through which access to and linkage of the data were enabled.<sup>10,11</sup> The randomization date into the trial was taken as time zero, and participants were followed up until 31 December 2019. Both the primary care data [Welsh Longitudinal General Practice (WLGP)] and population-scale national secondary care data [Patient Episode Database for Wales (PEDW)] were used to identify any diagnosis of AF, stroke (haemorrhagic or ischaemic), transient ischaemic attack (TIA), systemic embolism, and hospitalization for bleeds.<sup>12,13</sup> Prescription of anticoagulation medication was also identified from WLGP data. Death was identified from the Office for National Statistics (ONS) mortality data [Annual District Death Extract (ADDE)].<sup>14</sup>

#### Statistical analysis

The primary outcome for this study was AF diagnosis, and secondary outcome measures were stroke/TIA/SE (SSE) and death. Hospital admissions for acute coronary syndrome, venous thromboembolism, and bleeding were also assessed. Cox regression was used to provide both unadjusted and adjusted hazard ratios (HR) and 95% confidence intervals (CI) for clinical events and prescription of anticoagulation therapy during the follow-up period. Variables included in the adjusted model were age  $\geq$  75, sex, hypertension, diabetes, previous stroke, peripheral artery disease, and CHADS-VASc score  $\geq$  4. Kaplan–Meier plots were used to estimate event rates for AF diagnosis, SSE, and death between the groups. The CHADS-VASc score had been calculated for all patients in the main study, and the analysis, this variable was a significant predictor of AF. SPSS (version 28, IBM Corp., Armonk, NY, USA) and R (version 4.1.3) were used for analysis.

## Results

Successful linkage between study data and electronic health record (EHR) data sources within the SAIL Databank was achieved for 99.6% (997 of the original 1001 participants) in the study. The median period of follow-up was 4.2 years (1547 days, IQR 159), including the year of screening. *Table 1* shows the baseline characteristics of those participants included in this analysis by treatment group. There were no differences in the age and sex or medical history of the two groups with linked data, as observed in the original trial.

Over the full period of follow-up, although detection of AF was 38.7% greater in the iECG group compared with RC, this difference was not statistically significant (*Table 2*). There was no difference in the number of deaths, strokes, or TIA between the two groups.

Table 2	Number of	f clinica	l events and	l anticoagu	lation t	herapy	prescription

Outcome n (%)	iECG	Routine care	Unadjusted	Adjusted
Death	35 (7.0)	33 (6.6)	1.05 (0.65–1.69)	1.07 (0.66–1.73)
Atrial fibrillation	43 (8.6)	31 (6.2)	1.41 (0.89–2.24)	1.37 (0.86–2.19)
Stroke/TIA/SE	28 (5.6)	32 (6.4)	0.87 (0.52–1.44)	0.92 (0.54–1.54)
lschaemic stroke	23 (4.6)	27 (5.4)	0.84 (0.48–1.47)	0.90 (0.51–1.60)
Transient ischaemic attack	13 (2.6)	16 (3.2)	0.81 (0.39–1.68)	0.86 (0.41–1.82)
Acute coronary syndrome	10 (2.0)	21 (4.2)	0.47 (0.22–1.00)	0.45 (0.21–0.97)
Bleed	34 (6.8)	38 (7.6)	0.88 (0.56-1.40)	0.85 (0.53–1.35)
Anticoagulant therapy	50 (10.0)	37 (7.4)	1.38 (0.90–2.11)	1.28 (0.83–1.96)

Unadjusted and adjusted hazard ratios with routine care as the reference.

iECG, iPod ECG; SE, systemic embolism; TIA, transient ischaemic attack.

Although more participants in the iECG group were prescribed anticoagulant therapy, this difference did not reach statistical significance (*Table 2*). Of the 74 patients who developed AF, 66 were treated with anticoagulants (89%), and 8 were not (11%). The SAIL data governance protocol does not permit us to present absolute numbers due to the small number of patients (<5) not being prescribed anticoagulants in one or both groups. However, the proportion of participants who developed AF and were treated with anticoagulants was similar in the two groups (P = 0.71). Twenty-one participants received anticoagulant treatment for another indication.

Participants in the iECG care group were less likely to have an acute coronary syndrome event during the follow-up period than in the RC group (10 vs. 21, P = 0.044; *Table 2*).

There were no differences between the groups with regard to time to diagnosis of AF, stroke and SE, or death (Kaplan–Meier plots; *Figure* 1).

A subgroup analysis was undertaken in those with a CHADS-VASc score  $\geq 4$  at baseline, as this was an independent predictor for increased detection of AF with iECG screening. Similar numbers of diagnoses of AF, SSE, and deaths were seen in both groups, with fewer acute coronary syndromes occurring in the iECG group (*Table 3*), as observed in the overall study.

## Discussion

Although a 1-year period of iECG monitoring with the AliveCor Kardia Mobile device increased the detection of AF compared with RC during the screening period, there was no overall reduction in the number of strokes, TIA, systemic embolisms, or deaths over a mean follow-up period of 4.2 years from enrolment into the study. In addition, whilst the number of patients diagnosed with AF over the entire follow-up period, including the year of screening, was greater in the screening group, the difference between the groups no longer remained statistically significant over the longer-term follow-up. Hence, the initial advantage in detection of AF was gradually eroded beyond the screening period. Interestingly, patients in the iECG group were less likely to have acute coronary syndrome during follow-up. These findings were replicated when the analyses were restricted to only those with a higher likelihood of an AF diagnosis in the original screening study, who would also be expected to be at greater risk of events (CHADS-VASc score  $\geq$  4).

The original REHEARSE-AF study showed increased detection of AF (19 iECG vs. 5 RC, P = 0.007) and a trend towards fewer associated SSE (6 iECG vs. 10 RC, P = 0.34) events in the iECG group during 1 year of monitoring, suggesting that this might be a promising approach to reducing the risk of major complications of AF.<sup>5</sup> These early trends seen during the first year did not translate into a measurable reduction of hard clinical events in a cohort of this size. We note that the original study was not powered to evaluate clinical outcomes. The power of this follow-up is considerably improved with the four-fold increase in the follow-up time and hence the expected event count, but there is still considerable uncertainty in the estimates for clinical event differences, as reflected in the 95% CI for the HR. Thus we cannot exclude small effects on clinical events. We note that the HR point estimate for stroke is centred very close to 1 but with a 95% Cl from 0.54 to 1.54. However, as the regular iECG screening approach was discontinued after 1 year, the benefits of screening are only likely to have been realized by those patients in whom AF was diagnosed during the screening period. As such, this analysis cannot determine whether extending the period of screening beyond 1 year would result in further differential increases in AF diagnoses and fewer clinical events in the iECG compared with the RC group. Nonetheless, these findings suggest that the benefit of this screening approach may not be as marked as initially hoped for, if discontinued after 1 year. However, whether the initial promise of this approach to increase AF detection and reduce hard



**Figure 1** Kaplan–Meier plots describing time-to-event curves for diagnosis of atrial fibrillation (*A*), stroke/transient ischaemic at-tack/systemic embolism (*B*), and death (*C*). Shaded areas represent 95% confidence regions. Log-rank P = 0.28, 0.83, and 0.87 (Mantel–Cox), respectively. Day 0 on the *x*-axis is the point of randomization in the study.

clinical events could be realized by an ongoing screening strategy beyond 1 year would require further study in larger and/or longer clinical trials. Ongoing screening would also agree with the WHO principles of early disease detection, specifically that case finding should be a continuing process and not a 'once and for all project'.<sup>15</sup> However, the clinical and cost-effectiveness of such a strategy would need to be formally evaluated before being recommended. with a CHADS-VASc score > 4

Table 3	Number of clinical events and anticoagulation therapy prescription during the follow-up period in participant

Outcome n (%)	iECG ( <i>n</i> = 135)	Routine care $(n = 136)$	Unadjusted	Adjusted
Death	14 (10.4)	13 (9.6)	1.10 (0.52–2.33)	1.16 (0.53–2.53)
Atrial fibrillation	15 (11.1)	10 (7.4)	1.57 (0.71–3.50)	1.46 (0.65–3.27)
Stroke/TIA/SE	15 (11.1)	11 (8.1)	1.42 (0.65-3.10)	1.37 (0.61–3.09)
Ischaemic stroke	13 (9.6)	10 (7.4)	1.35 (0.59–3.07)	1.28 (0.54–3.04)
Transient ischaemic attack	6 (4.4)	<5ª	2.08 (0.52-8.33)	2.15 (0.53–8.67)
Acute coronary syndrome	<5 <sup>a</sup>	10 (7.4)	0.30 (0.08-1.10)	0.23 (0.06–0.83)
Bleed	13 (9.6)	11 (8.1)	1.21 (0.54–2.69)	1.10 (0.49–2.47)
Anticoagulant therapy	18 (13.3)	11 (8.1)	1.75 (0.82–3.70)	1.64 (0.77–3.48)

Unadjusted and adjusted hazard ratios with routine care as the reference.

<sup>a</sup>Governance restrictions within SAIL prohibit the reporting of numbers <5 due to privacy protection and disclosure control.

Two other randomized controlled trials have evaluated the impact of screening for AF on clinical outcomes, with differing results. In the LOOP study, a similar number of strokes and systemic embolisms were seen in patients aged 70-90 years without known AF but one risk factor for stroke undergoing monitoring with an implantable loop recorder (n = 1420) compared with controls (n = 4503) over a median follow-up period of 5.4 years.<sup>9</sup> However, in the recent STROKESTOP study, which followed 7165 patients who used a singlelead handheld device twice daily for 2 weeks and 14 381 controls for a median of 6.9 years, a reduction in the composite endpoint of any stroke/bleed/death (P = 0.045) was observed in the screening group but with no difference in the incidence of ischaemic stroke.<sup>8</sup> Notably, the difference in the composite endpoint emerged between the groups during the later period of follow-up (5–6 years), which was greater than in the current study. Furthermore, the patients in the STROKESTOP study were on average older (all aged 75-76 years on recruitment) than those in the REHEARSE-AF study [mean (SD) 72.6 (55) years] but also included those who already had a diagnosis of AF.

The VITAL-AF study evaluated 30 715 patients aged 65 years and over who attended primary care clinics over a period of 12 months.<sup>16</sup> Practices were randomized to offer iECG assessment with the AliveCor device during patient appointments vs. usual care. Although there were a similar number of diagnoses of AF and major adverse clinical outcomes in both study groups, this was effectively a study of opportunistic single-lead ECG testing in those attending primary care for usual reasons, rather than a trial of systematic screening for AF.

Considered together with the inconsistent results from these outcome studies, the evidence from our study suggests that the ultimate clinical impact of systematic screening for AF using single-lead ECG in at-risk populations remains uncertain and may not be as substantial as initially thought, at least with screening protocols of limited duration. However, we note that a number of larger outcome studies using different approaches in different target populations are ongoing, the results of which are eagerly awaited.<sup>17,18</sup>

Although these studies considered together do not currently support a role for systematic AF screening at present, they do not diminish the value of point-of-care single-lead ECG assessment as a clinical tool, especially in settings where a 12-lead ECG is not readily and/or rapidly available such as primary and community care settings or where they may be used as an alternative to ambulatory monitoring in appropriately selected patients with intermittent symptoms. Indeed, recent UK-based National Institute for Health and Care Excellence (NICE) guidelines have recently recommended the AliveCor for use in patients with suspected paroxysmal AF presenting with symptoms such as palpitations and who have been referred for ambulatory ECG monitoring by their clinician.  $^{19}\,$ 

The finding of fewer acute coronary syndromes in the iECG group was somewhat unexpected. It is possible that the nature of the monitoring process in this group led to increased interaction with their clinicians and better general preventive advice and treatment. However, a formal evaluation of these issues is outside the scope of this study and can only be speculative. Alternatively, it may also be due to a Type 1 error as this was not a primary outcome, given the relatively small number of events.

#### Limitations

The patients were predominantly of White European ethnicity and from a single UK regional health authority, which may limit the generalizability of the findings to other populations. Although the follow-up period is >4 years and a longer period may be required to fully evaluate the benefit of this screening approach and tease out potential clinical effects further, the fact that the gap in the proportion of patients with stroke or SE observed during the initial year of study (albeit nonsignificant) narrowed over the longer-term period of follow-up suggests that this would be unlikely to be the case.

The current guidelines do not discriminate between screen-detected vs. clinically detected AF regarding the recommended approach to management, including the antithrombotic strategy. However, there are fewer data available with regard to the thrombo-embolic risk and net clinical benefit of anticoagulation in screen-detected AF. Nonetheless, several studies have shown a greater risk of adverse outcomes in asymptomatic/screen-detected AF patients at increased cardiovascular disease risk.<sup>20–22</sup> Studies of patients with implanted devices have shown an increased risk of adverse events in patients with atrial high-rate episodes consistent with AF/atrial tachyarrhythmia, but these patients are not necessarily representative of the general population. Further randomized controlled trials are needed to determine the impact on outcome measures.

## Conclusions

In conclusion, 1-year of home-based, twice-weekly screening for AF did not lead to a reduction in cardiovascular-related events or all-cause death after a median 4.2-year follow-up compared with patients who had RC, even in those at highest risk of AF. Further research is required to identify a patient population and an ECG testing strategy that may obtain improved outcomes from a targeted screening programme for AF.

# Lead author biography



Elizabeth A. Ellins is a senior research officer based at Swansea University with an interest in cardiovascular disease prevention. She started her career as a vascular technologist specializing in non-invasive measures to assess vascular structure and function at the Institute of Child Health, University College London. She continued this at Cardiff University, where she obtained her PhD. Now at Swansea University, her interests have diversified into data science, using electronic health records to

investigate cardiovascular disease prevention with a particular interest in mental and women's health.

## **Author contributions**

J.P.H. and E.A.E. contributed to the conception of the work. E.A.E., D.E.H., M.B.G., and J.P.H. contributed to the analysis and interpretation of data for the work. E.A.E. drafted the manuscript. K.W., M.H., M.G., J.P.B., C.J.P., M.B.G., and J.P.H. were responsible for delivering the REHEARSE-AF clinical trial. K.W., D.E.H., A.A., M.H., M.G., J.P.B., C.J.P., M.B.G., and J.P.H. critically revised the manuscript. All authors gave final approval and agreed to be accountable for all aspects of the work ensuring integrity and accuracy.

### Data availability

The data used in this evaluation are openly available in the Secure Anonymised Information Linkage (SAIL) Databank at Swansea University, Swansea, UK.

### Acknowledgements

This study makes use of anonymized data held in the SAIL Databank. We would like to acknowledge all the data providers who make anonymized data available for research.

## Funding

The original study was funded by a joint grant from the Welsh Government Health Technology and Telehealth Fund and AliveCor Inc.

**Conflict of interest:** The original study was funded predominantly by the Welsh Government but in part by a project grant from AliveCor. The study data were analysed and reported independently without involvement of the company. J.P.H. had full access to all the study data and takes responsibility for its integrity and data analysis. None of the authors have received personal financial support for speaking or consulting on behalf of AliveCor Inc. There are no other disclosures to report.

### References

 Chugh SS, Havmoeller R, Narayanan K, Singh D, Rienstra M, Benjamin EJ, Gillum RF, Kim YH, McAnulty JH Jr, Zheng ZJ, Forouzanfar MH, Naghavi M, Mensah GA, Ezzati M, Murray CJL. Worldwide epidemiology of atrial fibrillation: a Global Burden of Disease 2010 Study. *Circulation*. 2014; **129**:837–847.

- Siontis KC, Gersh BJ, Killian JM, Noseworthy PA, McCabe P, Weston SA, Roger VL, Chamberlain AM. Typical, atypical, and asymptomatic presentations of new-onset atrial fibrillation in the community: characteristics and prognostic implications. *Heart Rhythm* 2016;**13**:1418–1424.
- Risk factors for stroke and efficacy of antithrombotic therapy in atrial fibrillation. Analysis of pooled data from five randomized controlled trials. Arch Intern Med 1994; 154:1449–1457.
- 4. Ruff CT, Giugliano RP, Braunwald E, Hoffman EB, Deenadayalu N, Ezekowitz MD, Camm AJ, Weitz JI, Lewis BS, Parkhomenko A, Lewis BS, Parkhomenko A, Yamashita T, Antman EM. Comparison of the efficacy and safety of new oral anticoagulants with warfarin in patients with atrial fibrillation: a meta-analysis of randomised trials. *Lancet* 2014;**383**:955–962.
- Halcox JPJ, Wareham K, Cardew A, Gilmore M, Barry JP, Phillips C, Gravenor MB. Assessment of remote heart rhythm sampling using the AliveCor heart monitor to screen for atrial fibrillation: the REHEARSE-AF study. *Circulation* 2017;**136**:1784–1794.
- Gladstone DJ, Wachter R, Schmalstieg-Bahr K, Quinn FR, Hummers E, Ivers N, Marsden T, Thornton A, Djuric A, Suerbaum J, von Grünhagen D, McIntyre WF, Benz AP, Wong JA, Merali F, Henein S, Nichol C, Connolly SJ, Healey JS. Screening for atrial fibrillation in the older population: a randomized clinical trial. JAMA Cardiology 2021;6:558–567.
- Steinhubl SR, Waalen J, Edwards AM, Ariniello LM, Mehta RR, Ebner GS, Carter C, Baca-Motes K, Felicione E, Sarich T, Topol EJ. Effect of a home-based wearable continuous ECG monitoring patch on detection of undiagnosed atrial fibrillation: the mSToPS randomized clinical trial. JAMA 2018;320:146–155.
- Svennberg E, Friberg L, Frykman V, Al-Khalili F, Engdahl J, Rosenqvist M. Clinical outcomes in systematic screening for atrial fibrillation (STROKESTOP): a multicentre, parallel group, unmasked, randomised controlled trial. *Lancet* 2021;**398**:1498–1506.
- Svendsen JH, Diederichsen SZ, Højberg S, Krieger DW, Graff C, Kronborg C, Olesen MS, Nielsen JB, Holst AG, Brandes A, Haugan KJ, Køber L. Implantable loop recorder detection of atrial fibrillation to prevent stroke (the LOOP study): a randomised controlled trial. *Lancet* 2021;**398**:1507–1516.
- Lyons RA, Jones KH, John G, Brooks CJ, Verplancke J-P, Ford DV, Brown G, Leake K. The SAIL Databank: linking multiple health and social care datasets. BMC Med Inform Decis Mak 2009;9:3.
- Ford DV, Jones KH, Verplancke J-P, Lyons RA, John G, Brown G, Brooks CJ, Thompson S, Bodger O, Couch T, Leake K. The SAIL Databank: building a national architecture for e-health research and evaluation. *BMC Health Serv Res* 2009;9:157.
- Primary Care GP dataset. https://saildatabank.com/saildata/sail-datasets/primary-caregp-dataset/. (23 June 2022).
- Patient Episode Database for Wales. http://www.wales.nhs.uk/document/176173. (23 June 2022).
- Annual District Death Extract. https://saildatabank.com/saildata/sail-datasets/annualdistrict-death-extract-adde/. (23 June 2022).
- Wilson JM, Junger GG. Principles and practice of screening for disease. In: Public Health Papers 34, . Geneva: WHO; 1968.
- Lubitz SA, Atlas SJ, Ashburner JM, Lipsanopoulos ATT, Borowsky LH, Guan W, Khurshid S, Ellinor PT, Chang Y, McManus DD, Singer DE. Screening for atrial fibrillation in older adults at primary care visits: VITAL-AF randomized controlled trial. *Circulation* 2022;**145**:946–954.
- Engdahl J, Svennberg E, Friberg L, Al-Khalili F, Frykman V, Kemp Gudmundsdottir K, Fredriksson T, Rosenqvist M. Stepwise mass screening for atrial fibrillation using N-terminal pro b-type natriuretic peptide: the STROKESTOP II study design. *EP Europace* 2017;**19**:297–302.
- Screening for atrial fibrillation with ECG to reduce stroke. https://doi.org/10.1186/ ISRCTN16939438. (July 1 2022).
- 19. National Institute for Health and Care Excellence. KardiaMobile for detecting atrial fibrillation. 2022.
- Zwartkruis VW, Geelhoed B, Suthahar N, Bakker SJL, Gansevoort RT, van Gelder IC, de Boer RA, Rienstra M. Atrial fibrillation detected at screening is not a benign condition: outcomes in screen-detected versus clinically detected atrial fibrillation. Results from the Prevention of Renal and Vascular End-stage Disease (PREVEND) study. *Open Heart* 2021;**8**:e001786.
- Boriani G, Laroche C, Diemberger I, Fantecchi E, Popescu MI, Rasmussen LH, Sinagra G, Petrescu L, Tavazzi L, Maggioni AP, Lip GYH. Asymptomatic atrial fibrillation: clinical correlates, management, and outcomes in the EORP-AF Pilot General Registry. *Am J Med* 2015;**128**:509–518.e502.
- 22. Gibbs H, Freedman B, Rosenqvist M, Virdone S, Mahmeed WA, Ambrosio G, Camm AJ, Jacobson B, Jerjes-Sanchez C, Kayani G, Oto Ali, Panchenko E, Ragy H, Kakkar AK. Clinical outcomes in asymptomatic and symptomatic atrial fibrillation presentations in GARFIELD-AF: implications for AF screening. *Am J Med* 2021;**134**:893–901.e811.